\begin{thebibliography}{10}

\bibitem{Agarwal91}
A.~Agarwal, D.~Chaiken, G.~D'Souza, K.~Johnson, D.~Kranz, J.~Kubiatowicz,
  K.~Kurihara, B.~H. Lim, G.~Maa, D.~Nussbaum, M.~Parkin, and D.~Yeung.
\newblock {The MIT Alewife Machine: A Large-Scale Distributed-Memory
  Multiprocessor}.
\newblock In {\em Proceedings of Workshop on Scalable Shared Memory
  Multiprocesors}. Kluwer Academic Publishers, 1991.

\bibitem{StarTNG94}
B.~S. Ang, Arvind, and D.~Chiou.
\newblock {{StarT the Next Generation: Integrating Global Caches and Dataflow
  Architecture}}.
\newblock In {\em Proceedings of the ISCA '92 Dataflow Workshop, Hamilton
  Island, Australia (to be published)}, May 1992.

\bibitem{CCDSM94}
B.~S. Ang, D.~Chiou, and Arvind.
\newblock {Issues in Building a Cache-Coherent Distributed Shared Memory
  Machine using Commercial SMPs}.
\newblock {CSG} Memo 365, {Laboratory for Computer Science, MIT}, Cambridge MA,
  December 1994.

\bibitem{arvind90}
Arvind and R.~S. Nikhil.
\newblock {Executing a Program on the {MIT} Tagged-Token Dataflow
  Architecture}.
\newblock {\em IEEE Transactions on Computers}, 39(3):300--318, March 1990.

\bibitem{BlumofeJoKuLeRaShZh94}
R.~D. Blumofe, C.~F. Joerg, B.~C. Kuszmaul, C.~E. Leiserson, K.~H. Randall,
  A.~Shaw, and Y.~Zhou.
\newblock {C}ilk: An efficient multithreaded runtime system.
\newblock Submitted for publication, December 1994.
\newblock Available via anonymous FTP from {\small\tt theory.lcs.mit.edu} in
  {\small\tt /pub/cilk/cilkpaper.ps.Z}.

\bibitem{Boughton94}
G.~A. Boughton.
\newblock {Arctic Routing Chip}.
\newblock In {\em {Parallel Computer Routing and Communication: Proceedings of
  the First International Workshop, PCRCW '94}}, volume 853 of {\em Lecture
  Notes in Computer Science}, pages 310--317. Springer-Verlag, May 1994.

\bibitem{Dally92}
W.~J. Dally and et~al.
\newblock {Architecture of a Message-Driven Processor}.
\newblock {\em IEEE Micro}, 12(2):23--39, 1992.

\bibitem{Dwarkadas93}
S.~Dwarkadas, P.~Keleher, A.~L. Cox, and W.~Zwaenepoel.
\newblock {Evaluation of Release Consistent Software Distributed Shared Memory
  on Emerging Network Technology}.
\newblock In {\em 4th. ACM Symposium on Principles and Practice of Parallel
  Programming (PPoPP)}. ACM, 1993.

\bibitem{Hagersten92}
E.~Hagersten.
\newblock {\em {Toward Scalable Cache Only Memory Architectures}}.
\newblock PhD thesis, The Royal Institute of Technology, Stockholm, Sweden,
  October 1992.

\bibitem{Heinrich94}
M.~Heinrich, J.~Kuskin, D.~Ofelt, J.~Heinlein, J.~Baxter, J.~P. Singh,
  R.~Simoni, K.~Charachorloo, D.~Nakashira, M.~Horowitz, A.~Gupta,
  M.~Rosenblum, and J.~Hennessy.
\newblock {The Performance Impact of Flexibility in the Stanford {FLASH}
  Multiprocessor}.
\newblock In {\em Proceedings of the Sixth International Conference on
  Architecture Support for Programming Languages and Operating Systems, San
  Jose, CA}, pages 274 -- 285, October 1994.

\bibitem{Joerg90}
C.~F. Joerg.
\newblock Design and implementation of a packet switched routing chip.
\newblock Technical Report MIT/LCS/TR-482, Laboratory of Computer Science,
  Massachusetts Institute of Technology, December 1990.

\bibitem{Keckler&Dally92}
S.~W. Keckler and W.~Dally.
\newblock {Processor Coupling: Integrating Compile Time and Runtime
  Scheduling}.
\newblock In {\em Proceedings of the 19th Annual International Symposium on
  Computer Architecture}, pages 202--213. ACM, 1992.
\newblock ISCA '92-Gold Coast, Australia.

\bibitem{Lenoski90}
D.~Lenoski, J.~Laudon, K.~Charachorloo, A.~Gupta, and J.~Hennessy.
\newblock {The Directory-Based Cache Coherence Protocol for the DASH
  Multiprocesor}.
\newblock In {\em Proceedings of The 17th Annual International Symposium on
  Computer Architecture, Seattle, WA}, pages 148--159, 1990.

\bibitem{Lenoski&others92}
D.~Lenoski, J.~Laudon, T.~Joe, D.~Nakahira, L.~Stevens, A.~Gupta, and
  J.~Hennessy.
\newblock {The DASH Prototype: Implementation and Performance}.
\newblock In {\em Proceedings of The 19th Annual International Symposium on
  Computer Architecture, Gold Coast, Australia}, pages 92--103, May 1992.

\bibitem{Li86}
K.~Li.
\newblock {\em Shared Virtual Memory on Loosely Coupled Multiprocessors}.
\newblock PhD thesis, Yale University, September 86.
\newblock (Also as YALE/DCS/RR-492).

\bibitem{Li89a}
K.~Li, M.~Stumm, D.~Wortman, and S.~Zhou.
\newblock {Shared Virtual Memory Accommodating Hetergeneity}.
\newblock CS-TR 210-89, Princeton University, Department of Computer Science,
  February 1989.

\bibitem{Id}
R.~S. Nikhil.
\newblock {Id Reference Manual, Version 90.1}.
\newblock {CSG} Memo 284-2, {Laboratory for Computer Science, MIT}, Cambridge
  MA, September 1990.

\bibitem{Nikhil94}
R.~S. Nikhil.
\newblock {Cid: A Parallel "Shared-memory" C for Distributed-memory Machines}.
\newblock In {\em {Proceedings of the 7th Annual Workshop on Languages and
  Compilers for Parallel Computing, Ithaca, NY}}, Lecture Notes in Computer
  Science. Springer-Verlag, August 1994.

\bibitem{Papadopoulos91}
G.~M. Papadopoulos.
\newblock {\em Implementation of a General-Purpose Dataflow Multiprocessor}.
\newblock The MIT Press, 1991.
\newblock Research Monograph in Parallel and Distributed Computing.

\bibitem{Papadopoulos93}
G.~M. Papadopoulos, G.~A. Boughton, R.~Greiner, and M.~J. Beckerle.
\newblock {*T: Integrated Building Blocks for Parallel Computing}.
\newblock In {\em Proceedings of Supercomputing '93, Portland, Oregon}, pages
  624--635, November 1993.

\bibitem{Reinhardt93}
S.~K. Reinhardt, M.~D. Hill, J.~R. Larus, A.~R. Lebeck, J.~C. Lewis, and D.~A.
  Wood.
\newblock {The Wisconson Wind Tunnel: Virtual Prototyping of Parallel
  Computers}.
\newblock In {\em ACM SIGMETRICS}, May 1993.

\bibitem{Reinhardt94}
S.~K. Reinhardt, J.~R. Larus, and D.~A. Wood.
\newblock {Tempest and Typhoon: User-Level Shared Memory}.
\newblock In {\em Proceedings of the 21st Annual International Symposium on
  Computer Architecture, Chicago, Il}, pages 325--336, April 1994.

\bibitem{Sakai89}
S.~Sakai, Y.~Yamaguchi, K.~Hiraki, Y.~Kodama, and T.~Yuba.
\newblock {An Architecture of a Dataflow Single Chip Processor}.
\newblock {\em Proceedings of the 16th Annual International Symposium on
  Computer Architecture, Jerusalem, Israel}, pages 46--53, 1989.

\end{thebibliography}

%% This section compares *T-ng with other work, both in terms of
%% architecture, and maybe performance for some of the most closely
%% related ones. 
%% 
%%   - its heritage as second incarnation of *T.
%%     * NIU much further away in *T-ng, and have lower bandwidth.  
%%     * addition of global shared memory support. 
%%   - FLASH
%%     * an earlier project then FLASH; much less aggressive shared
%%       memory support.  
%%     * We came from the message passing end and added shared memory.
%%       They came from the shared memory end and added message
%%       passing.   
%%     * *T-ng has faster path for purely local access, but suffers
%%       for local-global.  However, we may be able to recover some
%%       performance with page localization.  
%%     * compare some numbers 
%%   - DASH
%%     * Dash has no messaging support. 
%%     * Both use SMPs, but *T-ng is relying on software/firmware to
%%       run protocol.  Much less special hardware compared to Dash. 
%%     * incidentally, the hack to use higher order address bits to
%%       channel accesses to ACD is not required in DASH or FLASH. 
%%   - Alewife 
%%     * very different generation machine.  Alewife is very nice in
%%       many ways, but no longer possible to build that kind of
%%       machine with commercial microprocessors.  
%%     * a more realistic machine than Alewife: will run slightly
%%       modified OS. 
%%   - Typhoon 
%%     * got to find out more about this.  
%%   - Shrimp
%%     * got to find out more about this.  
%%   - Berkeley's NOW. 
%%     * much better performance!  
%% 
%%------------------------------------------------------------------------
\section{Related Work}
%%------------------------------------------------------------------------
\label{sec:related-wk}
%%------------------------------------------------------------------------


%% First consider parallel machines that influence the design of
%% *T-ng, or which are very similar to *T-ng in functionality.  

%% Next, let us consider other projects with the concept of Network of
%% Workstation idea.  

%% Monsoon, EM-4, J-Machine, M-Machine.
%% Alewife, Dash, Flash.
%% Wisconsin Wind Tunnel, NVM.

\start/ is heavily influenced by dataflow architectures.  Its message
passing architecture is optimized for low-latency messaging rather than
high-bandwidth messaging (although its bandwidth is very competitive.)
Though it is a more difficult objective to achieve, low-latency messaging
is essential for our compiler research.  Machines that have influenced us
in this area are the original *T project\cite{Papadopoulos93}, MIT's
Monsoon\cite{Papadopoulos91}, ETL's EM-4\cite{Sakai89}, the
J-Machine\cite{Dally92} and the M-Machine\cite{Keckler&Dally92}.

\start/'s software approach to cache-coherency is shared by other projects
as well.  The Wisconsin Wind Tunnel\cite{Reinhardt93} (WWT) uses minimal
hardware support\footnote{They hijack the ECC bits and handlers rather than
adding any additional hardware.  Unfortunately, this strategy cannot be
supported on more aggressive processor architectures which do not provide
precise ECC exceptions.} to implement shared memory.  Network Virtual
Memory\cite{Li89a,Li86} (NVM) takes advantage of virtual-memory management
hardware to maintain coherency on a page-granularity basis.

\start/ is remarkably similar in some ways to Typhoon\cite{Reinhardt94}, a
paper architecture developed at Wisconsin.  Typhoon, however, is not SMP
based and proposes a much larger degree of custom hardware for its message
passing and shared memory support than \start/ does.


Alewife\cite{Agarwal91} and FLASH\cite{Heinrich94} use varying degrees
of software in their coherency processing.  Alewife has hardware
support for maintaining coherence, but traps to software for
exceptional cases not supported in hardware.  Each Alewife site
consists of a modified SPARC 2 processor and a fully custom memory
controller, the CMMU.  Unlike \start/, Alewife cannot use standard
commercial software such as the operating system.

Cache-coherency on FLASH, like on \start/, is maintained completely in
software.  That software, however, runs on a special piece of
hardware, the Magic chip, which replaces the standard memory
controller.  The Magic chip is much more aggressive, technology-wise,
than \start/, and achieves better performance when a cache-miss
occurs, but requires much more design effort, not only in the special
hardware but in the system to use it.  FLASH's shared memory design is
conceptually cleaner since it avoids unnecessarily recrossing the L3
and thus eliminates some of the deadlock situations.

The Stanford DASH\cite{Lenoski&others92,Lenoski90} is similar to
\start/ in that it uses SMPs as building blocks for parallel machines.
It adds custom shared memory boards to provide cache-coherent shared
memory across multiple SMP sites.  Unlike \start/, all the protocol
processing is performed by hardware on the shared memory board.  All
the directory memory also resides on this board.  DASH's shared memory
implementation, unlike \start/'s, allows access to resident global
memory to proceed without going through the shared memory board,
unless coherency action has to be carried out on remote sites.
However, the protocol is fixed in hardware, and dedicated memory is
used to maintain directory information.

\start/'s approach to shared memory uses much simpler hardware than
any of the hardware supported shared memory schemes that we have
encountered, allows finer-grained coherency than NVM, and works with
much more aggressive processors than the Wisconsin Wind Tunnel.  We
believe that this system will allow us to conduct interesting research
into real problems using globally shared memory and running message
passing codes with very competitive performance.


%%------------------------------------------------------------------------
%%                               END
%%------------------------------------------------------------------------

